Author's response to reviews

Title: Bicultural identity among economical migrants from three south European countries living in Switzerland: Adaptation and validation of a new psychometric instrument

Authors:

Ariel Eytan (Ariel.Eytan@hcuge.ch)
Nuria Jene-Petschen (Nuria.Jene-Petschen@hcuge.ch)
Marianne Gex-Fabry (Marianne.Gex-Fabry@hcuge.ch)

Version: 4 Date: 8 May 2007

Author's response to reviews: see over
Authors: Ariel Eytan, Nuria Jene-Petschen and Marianne Gex-Fabry
Title: Bicultural identity among economical migrants from three south European countries living in Switzerland: Adaptation and validation of a new psychometric instrument
Journal: BMC Psychiatry
MS: 2077011396132068

To the BioMed Central Editorial Production Team:

Thank you for your message.

We made the requested changes

Formatting changes requested

Please ensure that all tables are cited in the main text in a numerically consecutive order starting with Table 1.

We removed the first reference to Table 3 on page 6 (It now reads “…Items are aimed at grasping everyday life, easily identifiable indicators of the individual’s cultural identity. The English equivalents (partially reproduced from the above mentioned original instrument by Cortés) are presented in this study. The French original version is available upon request...” Instead of “…The English equivalents (partially reproduced from the above mentioned original instrument by Cortés) of the 24 items are presented in Table 3. These items are aimed at grasping everyday life, easily identifiable indicators of the individual’s cultural identity. The French original version is available upon request…”

Indeed, the main purpose of Table 3 is to present component analysis.

We now have:

Table 1: "Scales for measuring cultural identity: from origins to a contemporary view" cited in the main text at the end of page 4.
Table 2: "Sample characteristics" cited p 9
Table 3: "Principal component analysis" cited p 10 line 8
Table 4: "Discriminant validity" cited p 10 line 19
Table 5: "Convergent validity is cited p 10 line 24

Tables are presented in correct order at the end of the text.

Best wishes,

Ariel Eytan.